Author's response to reviews

Title: A qualitative study of older adults' responses to sitting-time questions: do we get the information we want?

Authors:

Jannique GZ van Uffelen (jvanuffelen@hms.uq.edu.au)
Kristiann C Heesch (k.heesch@qut.edu.au)
Robert L Hill (hill.bob07@gmail.com)
Wendy J Brown (wbrown@hms.edu.au)

Version: 4 Date: 9 June 2011

Author's response to reviews: see over
Dear editor,

Please see our response to your request for a second set of changes to the format of our manuscript below (in capitals). The manuscript should now conform to journal style.

Kind regards,

Kristiann Heesch

==============
Major revisions (we require the author to make these changes)

References: references must be cited in numerical order. You have cited 27 between references 13 and 14 on page 4.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Structure: the 'Competing Interests' should go after the conclusion with the 'Author's Contributions' following it, then finally, 'Acknowledgements'.
WE HAVE REORDERED THESE SECTIONS.